18.06 Problem Set 6

Due at 4pm on Wednesday, October 26 in 2-106

Please PRINT your name and recitation information on your homework

1. Section 4.4, Problem 20 2. Section 4.4, Problem 21 3. Section 4.4, Problem 23 4. Section 4.4, Problem 34 5. Section 4.4, Problem 35 6. Section 5.1, Problem 3 7. Section 5.1, Problem 12 8. Section 5.1, Problem 13 9. Section 5.1, Problem 18 10. Section 5.1, Problem 19 11. Section 5.1, Problem 28 12. Section 5.1, Problem 34 13. Section 5.2, Problem 5 14. Section 5.2, Problem 17 15. Section 5.2, Problem 21 16. Section 5.2, Problem 25 17. Section 5.3, Problem 6 18. Section 5.3, Problem 7 More problems this week to give you practice on Chapters 4 and 5 before Exam 2. This is the last homework on orthogonality and determinants.